Author’s response to reviews

Title: A survey of the working status of family medicine physicians in clinics and hospitals in Korea

Authors:

Hyun-Young Shin (shydeborah@gmail.com; shy801117@gmail.com)

Version: 4 Date: 10 Apr 2020

Author’s response to reviews:

we corrected all the comments and confirmed all the figures are citied in the manuscript.